Check for updates

Citation: Janecek E, Waindok P, Bankstahl M, Strube C (2017) Abnormal neurobehaviour and impaired memory function as a consequence of *Toxocara canis*- as well as *Toxocara cati*-induced neurotoxocarosis. PLoS Negl Trop Dis 11(5): e0005594. <u>https://doi.org/10.1371/journal.</u> <u>pntd.0005594</u>

Editor: Ramesh Ratnappan, George Washington University, UNITED STATES

Received: January 13, 2017

Accepted: April 24, 2017

Published: May 8, 2017

Copyright: © 2017 Janecek et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: A PhD grant (grant no. S0229) of the Karl-Enigk-Foundation (<u>https://www.deutsches-</u> <u>stiftungszentrum.de/stiftungen/karl-enigk-stiftung</u>) for PW is gratefully acknowledged. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Abnormal neurobehaviour and impaired memory function as a consequence of *Toxocara canis*- as well as *Toxocara cati*induced neurotoxocarosis

Elisabeth Janecek^{1®}, Patrick Waindok^{1®}, Marion Bankstahl^{2,3}, Christina Strube¹*

1 Institute for Parasitology, Centre for Infection Medicine, University of Veterinary Medicine Hannover, Hanover, Germany, 2 Department of Pharmacology, Toxicology, and Pharmacy, University of Veterinary Medicine Hannover, Hanover, Germany, 3 Center for Systems Neuroscience, Hanover, Germany

So These authors contributed equally to this work.

* christina.strube@tiho-hannover.de

Abstract

Background

Neuroinvasive larvae of the worldwide occurring zoonotic roundworms *Toxocara canis* and *T. cati* may induce neurotoxocarosis (NT) in humans, provoking a variety of symptoms including cognitive deficits as well as neurological dysfunctions. An association with neuropsychological disorders has been discussed. Similar symptoms have been described in *T. canis*-infected mice, whereas data on *T. cati*-induced NT are rare. Therefore, it was aimed to obtain insights into the impact on neurobehaviour as well as progression of neurological symptoms and behavioural alterations during the course of NT directly comparing *T. canis*-and *T. cati*-infected mice as models for human NT.

Methodology/Principal findings

C57BL/6 mice were orally infected with 2000 embryonated *T. canis* or *T. cati* eggs, respectively, the control group received tap water. Mice were screened weekly for neurobehavioural alterations and memory function starting one day prior infection until 97 days post infection (pi; *T. canis*-infection) and day 118 pi (*T. cati*-infection, uninfected control). Mostly motoric and neurological parameters were affected in *T. canis*-infected mice starting day 20 pi with severe progression accompanied by stereotypical circling. In contrast, *T. cati*infected mice mostly showed reduced response to sudden sound stimulus (indicator for excitability) and flight behaviour starting day 6 pi. Interestingly, enhanced grooming behaviour was observed exclusively in *T. cati*-infected mice, indicating a possible role of neurotransmitter dysregulation. Reduced exploratory behaviour and memory impairment was observed in both infection groups with delayed onset and less severe progression in *T. cati*compared to *T. canis*-infected mice. **Competing interests:** The authors have declared that no competing interests exist.

Conclusions/Significance

Results highlight the need to consider *T. cati* beside *T. canis* as causative agent of human NT. Findings provide valuable hints towards differences in key regulatory mechanisms during *T. canis*- and *T. cati*-induced NT, contributing to a comprehensive picture and consequently a broader understanding of NT, which will aid in developing strategies towards prevention in addition to novel diagnostic and therapeutic approaches.

Author summary

The worldwide occurring zoonotic roundworms Toxocara canis and T. cati may cause the so-called neurotoxocarosis (NT) in humans, possibly resulting in a variety of behavioural alterations. Comparable alterations to those in humans have been described in T. canisinfected mice, however, reports on T. cati-induced NT are rare. Therefore, the main causative agent of human NT remains unknown. For a better understanding and more detailed characterization of T. canis- and T. cati-induced NT, infected and uninfected mice were exposed to several examinations regarding behavioural alterations and cognitive dysfunctions. Even though both roundworm species share many characteristics, pathogenicity as well as larval migration in the animal model has been described as quite different. The current study additionally highlights differences between behavioural alterations in infected mice. T. canis-infection mostly provoked neurological alterations, whereas T. cati-infection resulted in reduced excitability and flight-related parameters. Exploratory behaviour was reduced in both infection groups in addition to a negative effect of infection on memory function. The presented work provides a valuable basis for further studies and highlights the need for further investigations concerning consequences and differences between T. canis- and T. cati-induced NT.

Introduction

Toxocarosis, caused by infective third stage larvae (L3) of the worldwide occurring roundworms of dogs and cats, *Toxocara canis* and *Toxocara cati*, respectively, is one of the most common parasitic infections in humans. Nevertheless, the global importance of the zoonotic helminth is considered to be underestimated [1,2]. Consequently, toxocarosis has been listed as one of the five most neglected parasitic infections in humans targeted as priority for public health action by the Centers for Disease Control and Prevention (CDC) [3]. Infections are most frequently acquired by ingestion of embryonated infective eggs from the environment with poor sanitation as well as poor hygiene standards favouring the risk for infection. Thus, mostly children and people living in poverty are affected. Favorable environmental conditions for development of *Toxocara* eggs as well as transmission in tropical climates result in comparably high contamination as well as infection rates in these countries. Exemplarily, seroprevalences up to 92.8% in La Réunion, France, were reported, demonstrating frequent exposure of humans to *Toxocara*. Considerably lower seroprevalences are detected in temperate regions as for example 2.4% in Denmark [4].

Human toxocarosis is associated with several forms of disease with varying symptoms [1]. One of these forms, neurotoxocarosis (NT), is induced by neuroinvasive L3 and may result in pathological presentations such as meningitis, encephalitis and myelitis. Also, cerebral lesions

have been predominantly observed in the white matter in addition to occlusion of cerebral arterial vessels [5–7]. This CNS involvement may result in symptoms such as motor impairments, neuropsychological disturbances like dementia or depression as well as in behavioural alterations [7–13]. Previous studies also correlated *Toxocara*-seroprevalence with reduced cognitive development and function in children and young adults [14–17]. Sporadically, an implication in neurodegenerative diseases like multiple sclerosis and Alzheimer's disease has been suggested [18,19].

Similar observations on clinical symptoms and neurological alterations have been made in mice which are therefore considered suitable model hosts for human NT. However, an extensive neurotropism with persistence in the brain has solely been described for *T. canis* larvae. Contrary, T. cati larval migration to neuronal tissue is less frequently observed and larvae mainly accumulate in skeletal muscle tissue [20]. Both species cause structural damage such as malacia and demyelination, which is more severe in *T. canis*- than in *T. cati*-infected mice [21–24]. Additionally, previous studies demonstrated partially severe neurological symptoms as well as behavioural alterations and memory impairment in T. canis-infected mice [25-27], whereas behavioural data about T. cati-induced NT is scarce. However, previous studies indicate earlier onset accompanied by more severe neurological symptoms in T. canis- compared to T. cati-infected mice [24]. Nevertheless, dysregulations of genes associated with the biological modules "behaviour and taxis" as well as "sensory perception/neurological system process" have been described in T. canis- as well as in T. cati-infected mouse brains [28], indicating potential behavioural alterations caused by either Toxocara species. Based on the affinity to the CNS as well as observed symptoms, T. canis is considered the causative agent of most human cases of NT. Nevertheless, T. cati has previously been associated with human cases of NT and the infection risk should not be underestimated [12,29] as environmental contamination with T. cati eggs is considered to be high due to the uncontrolled defecation behaviour of cats [30,31].

Due to the high infection risk in combination with the impact of *Toxocara*-infection on neurological and behavioural processes, it was aimed to assess the impact on neurobehaviour as well as the progression of neurological symptoms during the course of NT directly comparing *T. canis*- and *T. cati*-infected mice as models for human NT. Obtained data will aid in further characterization of possible influences of infection on the paratenic host and give a closer insight into the pathogenesis of NT as data about neurological disorders and cognitive deficits due to toxocarosis are still rare [32].

Materials and methods

Ethic statement

Animal experiments were performed in accordance with the German Animal Welfare act in addition to national and international guidelines for animal welfare. Experiments were permitted by the ethics commission of the Institutional Animal Care and Use Committee (IACUC) of the German Lower Saxony State Office for Consumer Protection and Food Safety (Niedersächsisches Landesamt für Verbraucherschutz und Lebensmittelsicherheit) under reference numbers 33.9-42502-05-01A038 and 33.12-42502-04-15/1869.

Biological material and experimental infection

T. canis and *T. cati* eggs were obtained from faeces of experimentally infected dogs and cats, respectively, kept at the Institute for Parasitology, University of Veterinary Medicine Hannover, for continuous maintenance of *Toxocara* spp. strains (reference number 33.9-42502-05-

01A038). Eggs were purified by a combined sedimentation/flotation technique and allowed to embryonate for 4–5 weeks at 25°C with subsequent storage in tap water at 4°C until use.

C57BL/6JRccHsd (Harlan Laboratories, Horst, Netherlands) female mice were obtained at approximately 4 weeks of age. Mice were allowed 5 days of acclimatization followed by 8 days of training for behavioural assessments prior infection (see sections below). At 6 weeks of age, a total of 16 mice were orally infected with 2000 embryonated *T. canis* as well as 7 mice with 2000 *T. cati* eggs in a total volume of 0.5 ml tap water. The control group (n = 7) received tap water only. Maintenance of mice included a 12/12 hours dark/light cycle, standard rodent diet *ad libitum* as well as daily assessment of physical condition. Behavioural assessments were conducted weekly starting one day prior infection (-1 dpi) until 97 days post infection (dpi) for *T. canis*-infected mice or until day 118 pi for *T. cati*-infected and control mice (reference number 33.12-42502-04-15/1869). Differences in duration of trials are based on delayed clinical symptoms in *T. cati*-infected mice as well as on severe progression and resulting ethical concerns in *T. canis*-infected mice [24]. Brain sections of all infection groups were stained exemplary with haematoxylin and eosin following termination of respective trials to demonstrate brain infection and resulting structural damage in *Toxocara*-infected mice.

Physiological status and neurobehavioural phenotyping

Mice were tested weekly regarding their physiological status as well as neurobehavioural alterations based on a modified Irwin Screen protocol [33]. Acclimatization prior examination was allowed for 30 min in a designated testing room separate from the maintenance room. Assessment was conducted in two phases: The first phase included a total of 17 general observations regarding appearance, health and motoric function. Briefly, body weight of mice was recorded as well as their responsiveness to being lifted up by the tail. Additionally, tail suspension and arousal upon transferring mice from the scale to the observation cage was evaluated. Within the cage, body position, tail elevation, respiratory rate, skin colour, coat appearance, eyes (exophthalmos, ptosis as well as lacrimation) and salivation were recorded as parameters for physiological status. As indicators for neurological and motoric dysfunctions, pelvic elevation, leg position and tumbling motion of mice were recorded in addition to startle response in combination with the ability to hear by evaluating responsiveness/excitability of mice to a sudden sound stimulus. The second test phase was conducted after the assessment of general activity (see section below) and included handling of mice which comprised balance in a moving cage, inquisitiveness upon a presented object, escaping upon gentle touch, vibrissae reflex, exploration and placing behaviour, forelimb placing reflex, "vertical screen test" (ability to hold on to an upside down screen), righting reflex and general handling behaviour. Body temperature of mice was measured at the end of assessments. Regular behaviour was scored with "2", deviating behaviour was scored higher or lower according to a defined scoring system. Deviation of the group mean from the standard score was calculated as percentage deviation from the standard score, if more than 25% of mice showed the respective alteration. Detailed description of the individual testing parameters as well as the corresponding scoring system is provided in <u>S1 Appendix</u>. The experimental cage was cleaned with 0.1% acetic acid following each individual mouse assessment.

Assessment of general activity

General activity was assessed by observing activity of mice in the experimental cage for three minutes, recording the activity and position every 10 sec. Position was determined by subdividing the cage into 12 squares along the wall (7.7 cm x 7.8 cm) as well as the centre (7.7 cm x 23.4 cm). Activity was evaluated by determining the speed of movement (inability to move,

slow, normal and rapid) as well as the following parameters: walking, grooming, sitting and sitting in a corner, sniffing, rearing, rearing against the wall as well as urinating/defecation. Besides normal activities, stereotypic behaviour like circling or head flicking as well as seizures were noted. Detailed information about assessed activities is listed in <u>S2 Appendix</u>.

Assessment of sensorimotor function

Functional deficits after *Toxocara*-infection were evaluated based on the approach described by Bouet et al. [34]. An approximately 0.2 x 0.2 cm piece of tape was applied to each forepaw of mice with subsequent immediate release in the experimental cage. Time was recorded upon first contact for left and right paws separately as well as upon removal of the tape for each paw. If 150 sec were exceeded without removal, mice were classified unable to sense or remove the piece of tape and recorded with 150 sec for subsequent analyses. The experimental cage was cleaned with 0.1% acetic acid following each individual mouse assessment.

Assessment of memory function

To assess the effect of *Toxocara*-infection on memory function as well as potential differences between *T. canis-* and *T. cati-*induced NT, mice were conditioned in a classic maze to find a food reward starting 8 days prior infection (initial training). In addition to the weekly experimental evaluation, mice were allowed to find the food reward once a week (continuous training). Mice were deprived of food for 2 hours before the maze test, which was recorded by camera and subsequently analysed in terms of duration until discovery of food reward and entries into dead-end arms. Changes of direction were also considered as dead-end entries. Mice were allowed three trials during experimental procedure with individual mean times used for final analyses. Mice were removed from the maze when 360 sec without completing the task were exceeded which was used in analyses as the maximum duration. The maze measured 36 cm x 28 cm with a height of 15 cm and included a series of vertical walls without ceiling. Vertical walls were removable and the floor of the maze was covered with plastic coated covering paper, which was disposed and replaced after each individual run. Maze structure is provided in <u>S1 Fig</u>.

Statistical analyses

Obtained data was tested for statistical differences using an unpaired *t*-test with Welch's correction comparing the respective infection group to the control group each experimental day. If datasets did not pass normality tests, Mann-Whitney *U*-test was applied. Statistical analyses were conducted with GraphPad PrismTM software (version 6.03). The level of significance was set at $\alpha = 5\%$.

Results

General health observations

Severe progression of *T. canis*-infection was observed resulting in a total of 3 mice being euthanized prior termination of experiments due to poor general health. Additionally, two mice were euthanized within the *T. cati*-infection group before completing the assessments during the course of infection. Therefore, experiments were conducted with the following numbers of mice: control mice: n = 7 at day -1 to 118 pi; *T. canis*-infected mice: n = 16 at days -1 to 55 pi, n = 15 at day 62 pi, n = 14 at days 69 to 90 pi and n = 13 at day 97 pi; *T. cati*-infected mice: n = 7 mice at day -1 pi, n = 6 at days 6 to 111 pi and n = 5 at day 118 pi.



Fig 1. Histological sections of *Toxocara*-infected mouse brains. (A) *T. canis*- and (B) *T. cati*-infected mouse brain section. Solid arrows indicate *Toxocara* larvae and dashed arrow indicates accumulation of gitter cells.

https://doi.org/10.1371/journal.pntd.0005594.g001

Histological stains revealed cross sections of *Toxocara* larvae as well as vacuolization, meningitis, perivascular cuffs and haemosiderophages in *T. canis*- as well as in *T. cati*-infected mouse brains. Additionally, gitter cells were detected as a result of *T. canis*-induced NT. Exemplary sections are provided in Fig 1.

Regarding the body weight, *T. canis*-infected mice showed significantly higher body weights on day -1 and continuously lower weights than control mice starting day 27 pi. *T. cati*-infected mice also showed a significantly higher body weight day -1 and 13 pi, however, continuously lower body weights were not observed until day 62 pi. General progression of mean body weights of mice during the course of infection is provided in <u>S2 Fig</u>.

Body temperature of uninfected control mice was consistent throughout the experiment in contrast to *T. canis*-infected mice, which showed significantly lower body temperature starting day 69 pi. *T. cati*-infected mice partially showed significantly higher body temperatures during the course of infection, except for significantly lower body temperatures than control mice on day 118 pi. Determined body temperatures are mostly considered within physiological range with *Toxocara*-infected mice being at the upper or lower limit. Mean body temperatures of all experimental groups are provided in <u>S2 Fig</u>.

Neurobehavioural phenotyping

Overall, differences in general health as well as behaviour compared to uninfected controls (Fig 2) were observed for T. canis- and T. cati-infected mice. Both infection groups showed an increased respiratory rate [parameter (p) 6] as well as bend body position (p 4) with higher intensity in T. cati-infected mice during the course of infection. In T. canis-infected mice, neurological and motoric dysfunctions such as tumbling motions (Fig 3; p 17) and ataxia (p 15, 16) in addition to disturbed balance (p 18) and impaired righting reflexes (p 26) were observed. Day 27 pi, T. canis-infected mice started to show the inability to hold on to an inverted screen during the vertical screen test (p 25) and presented as less inquisitive (p 19) and explorative (p 22) during the chronic phase of infection (starting day 48 pi). Less inquisitive (p 19) and explorative (p 22) behaviour was also observed in T. cati-infected mice (Fig 4); however, the most striking feature were reduced reactions during fear- and flight-related assessments instead of neurological alterations. Particularly, assessments of excitability (p 1, 13) as well as escape parameters (p 1, 20) were affected. T. cati-infected mice additionally presented with ruffled as well as partially dirty coats (p 8) starting day 62 pi. Even though all three experimental groups showed habituation to being handled (p 27) during the course of infection as well as a reduced flight reaction when being touched with an object (p 20), these

alterations of behaviour were observed with higher intensity and more frequently in *Toxocara*infected mice. A detailed schematic overview of altered health as well as behavioural parameters is presented in Figs 2-4.

Assessment of general activity

Both infection groups showed differences in activity patterns during the course of infection compared to the uninfected control. *T. canis*-infected mice revealed most severe differences to control mice mainly in walking activity, which was significantly more frequently observed starting day 34 pi. During the acute phase of infection from days 6–20 pi, *T. canis*-infected mice were also sitting more frequently than control mice. Rearing on cage walls was significantly reduced during almost the entire course of infection. In contrast, grooming was significantly more frequently observed in *T. cati*-infected mice compared to control mice starting day 27 pi, whereas *T. canis*-infected mice showed increased grooming behaviour solely day 27 pi. As similarly described for *T. canis*-infected mice, *T. cati*-infected mice were more frequently walking than uninfected controls starting day 34 pi until day 104 pi. Even though more frequent sitting was only observed day 20 pi, *T. cati*-infected mice were moving significantly slower than uninfected controls starting day 20 pi. In contrast, *T. canis*-infected mice were not significantly slower than uninfected controls until day 62. Rearing as well as sniffing was irregularly significantly reduced



Fig 2. Altered health and behavioural parameters during the course of neurotoxocarosis in uninfected control mice. Arrows indicate if the parameter was increased or decreased in at least 25% of mice per group. Colours specify the severity of the presented alteration based on the group mean difference from the standard score (= 2), whereby colour categories represent the percentage deviation from the standard score. If less than 25% of mice per group showed behavioural alterations, no alteration was indicated.

https://doi.org/10.1371/journal.pntd.0005594.g002

00

70 00

00 07

40

FF 60

dave next infection

	days post mection	-1	0	13	20	21	34	41	40	55	02	09	70	03	90	91
General health observation	1. Reaction during tail fixation										_	^2	· · · · · · · · · · · · · · · · · · ·			\checkmark
	2. Hind limb tension during transfer							\checkmark	\checkmark	\downarrow	\downarrow	\downarrow	\checkmark	\checkmark	\checkmark	\checkmark
	3. Arousal upon transfer															\checkmark
	4. Body position		↑	1	1	↑	1	1	1	1	↑	↑	↑	1	\uparrow	\uparrow
	5. Tail elevation				-				\checkmark							
	6. Respiratory rate	1	\uparrow	\uparrow	\uparrow	↑	1	1	↑	\uparrow	1	1	1	\uparrow	1	1
	7. Skin colour / circulation		1				$\uparrow\downarrow$		¥	¢	\downarrow					
	8. Coat appearance			1				2			4 - 0	1				
	9. Exophthalmos							6- 						-		
	10. Ptosis															
	11. Lacrimation															
	12. Salivation													-		
	13. Sound response (excitability)	$\uparrow \downarrow$	1	↑	↑	1	1	↑	$\uparrow\downarrow$	$\uparrow\downarrow$	\downarrow	\checkmark	↑↓	\checkmark	\downarrow	\checkmark
	14. Hearing based on sound response							-								
	15. Pelvic position						\downarrow	\downarrow	4	\downarrow	1	4	\checkmark	\checkmark	\checkmark	\downarrow
	16. Position of legs				-			↑	1	↑	↑	↑	1	1	\uparrow	\uparrow
	17. Tumbling motion				↑	↑	1	↑	1	↑	↑	↑	↑	\uparrow	\uparrow	\uparrow
	18. Balance							\downarrow		\downarrow	\downarrow		≯	\downarrow	\checkmark	\checkmark
	19. Inquisitiveness				\downarrow		\downarrow		\checkmark	¢	\downarrow	\downarrow	\rightarrow	*	4	\checkmark
Handling	20. Touch-escape	\downarrow	1	\checkmark	\checkmark	\downarrow	\downarrow	\downarrow	\checkmark	4	\checkmark	↓	\checkmark	\checkmark	\checkmark	\downarrow
	21. Vibrissae reflex		-		_			↑						\rightarrow	\checkmark	\downarrow
	22. Exploration behaviour				↑			\downarrow	↑	\downarrow	\downarrow	\checkmark	→	\checkmark	\checkmark	\checkmark
	23. Postural reflex									\rightarrow		+	≯	→	\checkmark	\checkmark
	24. Forelimb placing reflex															
	25. Vertical screen test			\checkmark		\downarrow	\downarrow	\downarrow	\downarrow	\checkmark	\downarrow	\downarrow	\checkmark	→	\downarrow	1
	26. Righting reflex											*	\checkmark	\checkmark	\checkmark	\checkmark
	27. Handling behaviour		\downarrow	*	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\checkmark	\downarrow	\downarrow	\downarrow	\downarrow	1
								21-40)% 4	1-60%	6 61	-80%	81-	100%		

20

Fig 3. Altered health and behavioural parameters during the course of *Toxocara canis*-induced neurotoxocarosis. Arrows indicate if the parameter was increased or decreased in at least 25% of mice per group. Colours specify the severity of the presented alteration based on the group mean difference from the standard score (= 2), whereby colour categories represent the percentage deviation from the standard score. If less than 25% of mice per group showed behavioural alterations, no alteration was indicated.

https://doi.org/10.1371/journal.pntd.0005594.g003

in both infection groups over the course of infection. Unusual behaviour such as spatial disorientation and stereotypical circling was exclusively observed in *T. canis*-infected mice, starting day 27 pi and 34 pi, respectively. Urination/defecation as well as corner sitting was rare (<5%) and no preference for any position within the cage was detected in all experimental groups. An overview of most commonly conducted activities is provided in Fig 5.

Assessment of sensorimotor function

Toxocara-infected mice showed altered sensory as well as motor function over the course of experimental infection with *T. canis*-infected mice showing more impaired reactions than *T. cati*-infected mice. Initial significantly increased time-to-contact of infected mice may be attributed to unsatisfactory adaptation to respective handling prior infection. During progression of infection, *T. canis*-infected mice showed significantly increased values days 41 and 48 pi as well as the remaining five examination days 69–97 pi. Similarly, *T. cati*-infected mice showed significantly increased time-to-contact compared to control mice as of day 76–118 pi with exception of day 111 pi. Consistent significantly longer times for removing the piece of

13 20 27 34 41 48 55 62 69 76 83 90 97 104 111 118

days post infection

	-	_								_	_	_		_	_	_	_		_
ealth observation	1. Reaction during tail fixation				+		$\mathbf{\downarrow}$	\downarrow	\downarrow	\downarrow	\downarrow	\checkmark	\checkmark	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow
	2. Hind limb tension during transfer								\checkmark	\checkmark	\downarrow	\checkmark	\checkmark	\downarrow	\downarrow	\downarrow	\checkmark	\checkmark	\downarrow
	3. Arousal upon transfer										2						¥		
	4. Body position		↑	↑	1	↑	↑	↑	↑	↑	↑	↑	↑	↑	1	↑	↑	1	↑
	5. Tail elevation			4											1		1	-	
	6. Respiratory rate		↑	↑	↑	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	\uparrow	1	\uparrow	\uparrow
	7. Skin colour / circulation		\downarrow	↑														1	
	8. Coat appearance		↑	↑							→	1	♦	↑	→	→	1	1	1
	9. Exophthalmos																		
	10. Ptosis					↑							↑			→			↑
드	11. Lacrimation									1	↑		1		↑	Ť		\uparrow	↑
Genera	12. Salivation				· · · · · ·										Ū,				
	13. Sound response (excitability)		↓	\downarrow	\downarrow	\downarrow	\mathbf{V}	¢	≁	*	\leftarrow	\checkmark	\downarrow	\downarrow	$\mathbf{+}$	$\mathbf{+}$	$\mathbf{+}$	→	\downarrow
	14. Hearing based on sound response									5		5 0	1 - N						
	15. Pelvic position												+					+	
	16. Position of legs										\leftarrow	\checkmark		¥	¢	¢	Ł	\rightarrow	\downarrow
	17. Tumbling motion		↑		↑		↑				↑	↑		↑	→	→	1	1	↑
Handling	18. Balance																1	\downarrow	Î
	19. Inquisitiveness	1	↑	↑		_	1				_	+		\downarrow	¥	≁	↓	\downarrow	\checkmark
	20. Touch-escape		↑	\checkmark	+	*	+	↓	*	↓	\downarrow	$\mathbf{+}$	+	↓	↓	4	4	\downarrow	↓
	21. Vibrissae reflex																		
	22. Exploration behaviour									\checkmark	\downarrow	\checkmark	\downarrow	4	\downarrow	\downarrow	↓	4	\downarrow
	23. Postural reflex																		
	24. Forelimb placing reflex				1						() (
	25. Vertical screen test											1			\downarrow	\downarrow	\downarrow	()	
	26. Righting reflex				\downarrow											\downarrow		\downarrow	
	27. Handling behaviour		¥	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	\downarrow	1	\downarrow	\downarrow	\downarrow	1	\downarrow	\downarrow	\downarrow
							20%	21-	40%	41-	60%	61-	80%	81-1	00%				

-1 6



https://doi.org/10.1371/journal.pntd.0005594.g004

tape from their paw were observed days 41–97 pi for *T. canis*-infected mice. Contrary, *T. cati*-infected mice only showed significantly increased time-to-remove days 83 and 90 pi. Detailed information about time-to-contact and time-to-remove are provided in Fig <u>6</u>.

Assessment of memory function

T. canis-infected mice required significantly more time to find the food reward than control mice starting day 27 pi. The mean duration upon finding the food reward increased over time, which was also observed regarding the number of entries into dead-end arms, starting to be significant as of day 20 pi. *T. cati*-infected mice required significantly more time than control mice 4 weeks later than *T. canis*-infected mice (starting day 55 pi), but did not show continuous entries into dead-end arms. Significantly more dead-end arms than control mice were solely entered on day 62 pi as well as days 83–97 pi. Detailed progression of impaired memory functions is provided in Fig 7.

Discussion

Toxocara-induced NT may result in a variety of neurological symptoms such as behavioural alterations as well as neurological disturbances in paratenic hosts, including humans. To date,



Fig 5. Alterations in activity patterns during the course of neurotoxocarosis. (A) uninfected control mice, (B) *T. canis*-infected mice, (C) *T. cati*-infected mice. Individual bars represent 100% of recorded activity with specific activities presented as respective fractions. Asterisks indicate significant differences in activities of the respective infection group compared to the uninfected controls. Note that intense reduction of "rearing on wall" and "rearing" in both infection groups did not allow presentation of asterisks in respective bars. Significantly reduced rearing on wall was additionally observed days 83–97 pi in *T. canis*-infected mice (B) and day 118 pi in *T. cati*-infected mice (C). Also, significantly reduced rearing was observed in *T. canis*-infected mice (B) days 6–20 pi and 34–97 pi days as well as in *T. cati*-infected mice (C) days 27–48 pi and 90–97 pi.

https://doi.org/10.1371/journal.pntd.0005594.g005





Fig 6. Assessment of sensorimotor function in *Toxocara canis*- and *T. cati*-infected mice. (A) Time-to-contact. Time required for mice to notice piece of tape on their front paws. (B) Time-to-remove. Time required for mice to remove piece of tape from their front paws. Ends of boxes define the 25th and 75th percentiles, with a line at the median and error bars defining the 10th and 90th percentiles. Asterisks indicate significant differences of the respective infection group compared to the uninfected control. * $p \le 0.05$; ** $p \le 0.001$; ** $p \le 0.001$.

https://doi.org/10.1371/journal.pntd.0005594.g006

most studies are focusing on the influence of *T. canis* as the presumably more frequent causative agent of human toxocarosis. However, diagnosis remains challenging and the potential infection risk with *T. cati* larvae may be underestimated as it is generally considered high due



Fig 7. Assessment of memory functions in *Toxocara canis*- and *T. cati*-infected mice using the classic maze. A) Mean time required for finding food reward. B) Number of entries into dead-end arms. Ends of boxes define the 25th and 75th percentiles, with a line at the median and error bars defining the 10th and 90th percentiles. Asterisks indicate significant differences of the respective infection group compared to the uninfected control. $*p \le 0.05$; $**p \le 0.001$; $**p \le 0.001$.

https://doi.org/10.1371/journal.pntd.0005594.g007

to uncontrolled defecation habits of cats in public places like playground sandpits $[\underline{29}-\underline{31}]$. Nevertheless, studies regarding influence of *T. cati* larvae on behaviour and neurological

functions as well as the direct comparison of *T. canis-* and *T. cati-* induced NT in terms of behavioural alterations and influence on memory function are rare. The presented study aids in filling the gap to provide a comprehensive picture of *T. canis-* and *T. cati-* induced NT and its progression in the mouse as model for human NT.

Both infection groups showed differences in general health and neurobehavioural parameters compared to uninfected mice, demonstrating the need to consider *T. cati* as a causative agent of human NT. Nevertheless, more severe pathogenesis in *T. canis*- than in *T. cati*infected mice was observed. Alterations were mostly detected during the chronic phase of infection with effects of *T. cati*-infection appearing delayed compared to *T. canis*-infection. These phenotypic results considerably complement data regarding more severe progression in *T. canis*-infection, with concurrent delayed onset in *T. cati*-infection in terms of clinical symptoms as well as transcriptional and histopathological alterations in the brain [22,24,35,36]. In contrast to the presented study, previously conducted behavioural studies showed no differences in general health of *T. canis*-infected mice compared to uninfected controls, but behavioural differences were observed [37], possibly indicating adaptive manipulation. Differences to presented data may depend on different mouse strains as well as infection doses as these factors influence progression of infection [21,35,38,39]. These factors also demonstrate the importance of directly comparing effects of *T. canis*- and *T. cati*-induced NT under identical conditions to thouroughly characterize effects of the parasites on the host.

In T. canis-infected mice, most parameters associated with neurological dysfunction were deviant from behaviour of uninfected controls. Tumbling motions were well observed starting day 20 pi indicating an early onset of neurological involvement due to early migration of larvae to the brain. Less severe neurological dysfunctions in T. cati-infected mice may be attributed to generally lower larval numbers in the brain, even though early migration of T. cati larvae to the brain has also been observed [21,22,26]. Circling as stereotypical behaviour was solely detected in T. canis-infected mice supporting the observation of severe neurological involvement. Even though mentioned neurological dysfunctions result in a selective advantage regarding uptake by the definitive host, behavioural alterations are most likely attributed to side-effects of pathology-rather than adaptive manipulation-and may be coincidentally beneficial for the parasite [38,40]. This is also supported by increased excitability levels (indicated by response to a sudden sound stimulus) in T. canis-infected mice until day 41 pi, which is not in accordance with adaptive manipulation as this behaviour would prevent exposure of infected mice in the environment. By contrast, T. cati-infection mostly provoked reduced excitability as well as flight reaction when being picked up by the tail resulting in increased probability of ingestion by predators within the environment. In addition to a milder course of infection, neurological disturbances like tumbling motions were observed only sporadically leading to the assumption of a better adaptation of T. cati larvae to the paratenic host. Therefore, contrary to T. canis-induced NT, T. cati-induced behavioural alterations are likely to be attributed to adaptive manipulation to facilitate transmission of larvae. Reasons for this clearly demonstrated deviant influence of the two Toxocara species remain unclear. It may be speculated that lower larval numbers in addition to less severe pathology in *T. cati*-infected mice provokes the parasite to induce alternative strategies to facilitate transmission of larvae from the paratenic to the definitive host. Reduced defence during physical handling was most intensely observed in T. cati- and to a lesser extent in T. canis-infected mice. In T. cati-infected mice, this observation might be attributed to generally reduced flight behaviour; whereas reduced defence in T. canis-infected mice is most likely a result of described pathology. Reduced defence in combination with reduced inquisitiveness and exploration in both Toxocarainfected groups are in accordance with results of activity assessments in terms of reduced rearing and increased walking. Tendency of incuriosity about their environment as well as high

activity levels (walking) in familiar or novel areas results in higher vulnerability to predators [41,42]; however, it remains unclear if this behaviour is attributed to pathology or adaptive manipulation. Even though suggested host manipulation is not beneficial for *Toxocara* spp. within the human host, pathology accompanied by behavioural as well as neurological alterations such as confusion, ataxia or dementia severely influences quality of life of affected patients [32]. Wether human *Toxocara*-infections result in personality changes as well as delayed reaction times, increasing the risk of e.g. accidents as described for human *Toxoplasma*-infection [43,44], has not been investigated yet.

Interestingly, the extensive grooming behaviour in *T. cati*-infected mice during activity assays was observed prior to ruffled, partially dirty, appearance during the chronic phase of infection (day 27 pi vs. day 62 pi). Ruffled appearance may be connected to effects of larval migration on general health as T. cati-infected mice also showed more severe deviations in body position compared to T. canis-infected mice, possibly indicating persistence of T. cati-larvae and pathological involvement throughout the skeletal muscle [22,35]. This persistence could also affect muscle function, nevertheless, T. cati-infected mice only showed sporadic inability to hold on to the vertical screen, whereas T. canis-infected mice showed marked inabilities during the course of infection, possibly attributed to reduced muscle strength [45,46] as well as poor general health. Furthermore, ruffled appearance of T. cati-infected mice and especially enhanced grooming may result from disturbances in hormone and neurotransmitter levels as these are inducing factors for altered self-grooming behaviour in rodents [47– 49]. Self-grooming has therefore been discussed as an indicator for repetitive behaviour and consequently as potential model for human psychiatric disorders [48]. Disturbed levels of neurotransmitters e.g. depressed levels of dopamine are implicated in behavioural alterations in a variety of psychiatric diseases like schizophrenia and depression [50,51]. A potential correlation between Toxocara-seropositivity and neuropsychiatric disorders like schizophrenia, seizures and cognitive deficits in human patients has been discussed [32]. Reduced dopamine as well as serotonin and GABA levels in T. canis-infected mouse brains [50] highlight the need for a more detailed characterization of NT to confirm potential correlation. As a variety of neurotransmitters induce a reduction of rearing behaviour and also modulate stress- and anxiety-related behaviour in rodents [48,49], recorded behavioural alterations may also be attributed to respective dysregulations. However, contrary to T. canis-infection, altered neurotransmitter levels during T. cati-induced NT have not been investigated yet. It may be speculated that T. cati-infection induces a different pattern of neurotransmitter dysregulation, which may account for observed marked increased grooming and reduced flight behaviour. Different neurotransmitter dysregulations may therefore be hypothesised as a key factor for differences in behavioural alterations during T. canis- and T. cati-induced NT.

Sensorimotor assessment indicated reduced sensation on paws, which was also observed earlier in *T. canis*- than in *T. cati*-infected mice with motoric impairment being more extensive in *T. canis*-infected mice. As some of the core functions of the cerebellum are coordination and balance, structural damage influences those functions. Recorded motor incoordination as well as ataxia may therefore be assigned to the pathological alterations in cerebellar tissue observed in previous studies [21,52,53]. Less severe structural damage during *T. cati-* in comparison to *T. canis*-induced NT [22] may contribute to less severe impairment in *T. cati-* infected mice. The altered sensorimotor behaviour in both infection groups may particularly be attributed to occurring axonal damage which is associated with neuromotor and sensory dysfunction [22,24,54]. Motor impairment as well as sensory impairment and dysesthesia in *Toxocara*-positive humans [12,55–57] may therefore also be attributed to axonal damage, particularly requiring further investigations. As mice predominantly sense the strip on their paw even if delayed, reduced removal may be an effect of impaired balance of *T. canis*-infected

mice. In contrast, prolonged removal times of T. cati-infected mice may not result from motoric impairment in terms of balance, but rather the neurological phenomenon "slower motion" already recorded during activity assays. The general slower motion in T. cati-infected mice also contributes most likely to significantly longer duration in the maze before finding the food reward compared to uninfected controls. Nevertheless, maze results show a significant effect on memory function in both infection groups. To our knowledge, this is the first report on T. cati larvae influencing memory function of the paratenic host. Again, T. canisinfected mice showed an earlier onset of impaired memory function as several attempts to fulfil the task starting day 20 pi were required. T. cati-infection had a less pronounced effect on memory and condition may have been transient as infected mice were not consistently requiring more dead-end entries than uninfected controls during the chronic NT phase as was the case with T. canis-infected mice. Transient behavioural alterations have been described in Toxocara- as well as in Toxoplasma-infected rodents, nevertheless, strains and infection doses were not comparable to the present study, which is an influential factor in disease progression [21,35,46,58]. Less pronounced alterations in *T. cati*-infected mice may be on account of fewer larvae in the brain, as Toxocara larval or Toxoplasma cysts numbers correlate with the degree of behavioural alteration [22,26,59]. Memory impairment in rodents may strongly influence survival factors like recalling resources and use of visual cues. As in the presented study, similar experiments provided evidence of memory impairment as T. canis-infected mice required significantly longer times than uninfected controls to locate vital resources like water in a given area after a preceded deprivation period [26]. Cerebellar lesions and damage in Toxocara-infected mice may contribute to impairments as parts of the cerebral cortex involved in cognitive regulation are connected to the cerebellum. Therefore, cerebellar integrity is required for cognitive functions [54] and may be influenced by the infection. Suspected neurodegeneration in *Toxocara*-infected mice [28] may particularly influence observed memory dysfunction as neurodegenerative cerebellar diseases in humans have been shown to influence explicit memory without affecting implicit memory [24,32,52]. Therefore, results obtained in mice may be comparable to observed memory impairment of *Toxocara*-positive human patients [7,10,11,13] as explicit memory, an intentional process to recall facts and experiences, is required to recall the location of the food reward. It may therefore be speculated that neurodegeneration constitutes an important part in human NT. Additionally, particular brain regions may be affected by larval persistence and adverse immune reactions. Damage of e.g. the hippocampus, which is implicated in development and maintenance of spatial learning as well as reduced exploratory behaviour [54], may result in recorded alterations. However, a detailed characterization of brain areas involved during NT is not yet available. Such investigations are desirable to better comprehend behavioural or neurological alterations in the infected host. Correlation of learning impairment and Toxocara-seropositivity in children additionally highlights the need for further detailed characterization of NT [15,17].

The current results clearly show that *T. cati*-induced NT also influences behavioural parameters, even though to a lesser extent than *T. canis*-induced NT. It still may be assumed that neurological disturbances such as ataxia and balance incoordination in humans may mostly be attributed to the—extrapolated from the mouse model—likely more severe *T. canis*-infection. However, based on obtained results, memory impairment as well as cognitive dysfunction frequently observed in NT patients [7,10,11] may be induced by either *Toxocara* species. *T. cati* should therefore strongly be considered as a causative agent of human NT as well. Overall, the conducted study evidently demonstrates an earlier onset of severe neurological and behavioural alterations in *T. canis*-infected mice with more severe progression of disease than in *T. cati*-infected mice. Observed differences in behavioural and neurological alterations provide valuable hints towards varying key regulatory mechanisms during *T. canis*- and *T. cati*-induced NT,

contributing to a broader understanding of NT, which will aid in determining targets for prevention as well as novel diagnostic and therapeutic approaches.

Supporting information

S1 Appendix. Physiological status and neurobehavioural phenotyping. Parameters and scoring modified according to Irwin (1968). (DOCX)

S2 Appendix. Assessment of general activity. Parameters and scoring modified according to Irwin (1968).

(DOCX)

S1 Fig. Classic maze for assessment of memory function. Schematic overview of the maze used in the presented study. The star indicates the starting position of mice and the flag the position of the food reward. (TIF)

S2 Fig. Mean body weights and temperatures of uninfected controls and *Toxocara*-infected mice during the course of infection. A) Body weight and B) body temperature. Ends of boxes define the 25th and 75th percentiles, with a line at the median and error bars defining the 10th and 90th percentiles. Asterisks indicate significant differences of the respective infection group compared to the uninfected control. (TIF)

Acknowledgments

The authors thank Mathias Boelke for technical assistance during behavioural testing.

Author Contributions

Conceptualization: CS.

Formal analysis: EJ PW.

Funding acquisition: PW CS.

Investigation: EJ PW.

Methodology: MB.

Project administration: CS.

Supervision: CS.

Visualization: EJ PW.

Writing - original draft: EJ.

Writing - review & editing: CS PW MB.

References

- Magnaval JF, Glickman LT, Dorchies P, Morassin B. Highlights of human toxocariasis. Korean J Parasitol. 2001; 39: 1–11. <u>https://doi.org/10.3347/kjp.2001.39.1.1</u> PMID: <u>11301585</u>
- Hotez PJ, Wilkins PP. Toxocariasis: America's most common neglected infection of poverty and a helminthiasis of global importance? PLoS Negl Trop Dis. 2009; 3: e400. <u>https://doi.org/10.1371/journal.pntd.0000400</u> PMID: <u>19333373</u>

- Toxocariasis. Centers for Disease Control and Prevention. 2013. <u>https://www.cdc.gov/parasites/npi/</u>. Accessed 2016 November 01
- Macpherson CN. The epidemiology and public health importance of toxocariasis: A zoonosis of global importance. Int J Parasitol. 2013; 43: 999–1008. <u>https://doi.org/10.1016/j.ijpara.2013.07.004</u> PMID: 23954435
- Sommer C, Ringelstein EB, Biniek R, Glockner WM (1994) Adult *Toxocara canis* encephalitis. J Neurol Neurosurg Psychiatry. 1994; 57: 229–231. PMID: <u>8126514</u>
- Xinou E, Lefkopoulos A, Gelagoti M, Drevelegas A, Diakou A, Milonas I, et al. CT and MR imaging findings in cerebral toxocaral disease. AJNR Am J Neuroradiol. 2003; 24: 714–718. PMID: <u>12695211</u>
- Moreira-Silva SF, Rodrigues MG, Pimenta JL, Gomes CP, Freire LH, Pereira F. Toxocariasis of the central nervous system: with report of two cases. Rev Soc Bras Med Trop. 2004; 37: 169–174. PMID: <u>15094904</u>
- Finsterer J, Auer H. Neurotoxocarosis. Rev Inst Med Trop Sao Paulo. 2007; 49: 279–287. PMID: 18026633
- Hill IR, Denham DA, Scholtz CL. *Toxocara canis* larvae in the brain of a British child. Trans R Soc Trop Med Hyg. 1985; 79: 351–354. PMID: <u>4035735</u>
- Richartz E, Buchkremer G. Cerebral toxocariasis: a rare cause of cognitive disorders. A contribution to differential dementia diagnosis. Nervenarzt. 2002; 73: 458–462. PMID: <u>12078025</u>
- Scheid R, Jentzsch RT, Schroeter ML. Cognitive dysfunction, urinary retention, and a lesion in the thalamus—beware of possible toxocariasis of the central nervous system. Clin Neurol Neurosurg. 2008; 110: 1054–1057. https://doi.org/10.1016/j.clineuro.2008.06.014 PMID: 18687519
- Fukae J, Kawanabe T, Akao N, Kado M, Tokoro M, Yokohama K, et al. Longitudinal myelitis caused by visceral larva migrans associated with *Toxocara cati* infection: case report. Clin Neurol Neurosurg. 2012; 114: 1091–1094. https://doi.org/10.1016/j.clineuro.2012.02.026 PMID: 22424728
- Eberhardt O, Bialek R, Nagele T, Dichgans J. Eosinophilic meningomyelitis in toxocariasis: case report and review of the literature. Clin Neurol Neurosurg. 2005; 107: 432–438. <u>https://doi.org/10.1016/j.</u> clineuro.2004.10.003 PMID: 16023542
- Marmor M, Glickman L, Shofer F, Faich LA, Rosenberg C, Cornblatt B, et al. *Toxocara canis* infection of children: epidemiologic and neuropsychologic findings. Am J Public Health. 1987; 77: 554–559. PMID: 3565646
- Walsh MG, Haseeb MA. Reduced cognitive function in children with toxocariasis in a nationally representative sample of the United States. Int J Parasitol. 2012; 42: 1159–1163. <u>https://doi.org/10.1016/j. ijpara.2012.10.002</u> PMID: 23123274
- Erickson LD, Gale SD, Berrett A, Brown BL, Hedges DW. Association between toxocariasis and cognitive function in young to middle-aged adults. Folia Parasitol (Praha). 2015; 62: 048.
- Nelson S, Greene T, Ernhart CB. *Toxocara canis* infection in preschool age children: risk factors and the cognitive development of preschool children. Neurotoxicol Teratol. 1996; 18: 167–174. PMID: 8709928
- Liao CW, Fan CK, Kao TC, Ji DD, Su KE, Lin YH, et al. Brain injury-associated biomarkers of TGFbeta1, S100B, GFAP, NF-L, tTG, AbetaPP, and tau were concomitantly enhanced and the UPS was impaired during acute brain injury caused by *Toxocara canis* in mice. BMC Infect Dis. 2008; 8: 84. https://doi.org/10.1186/1471-2334-8-84 PMID: 18573219
- Söndergaard HP, Theorell T. A putative role for *Toxocara* species in the aetiology of multiple sclerosis. Med Hypotheses. 2004; 63: 59–61. <u>https://doi.org/10.1016/j.mehy.2004.02.014</u> PMID: <u>15193348</u>
- Strube C, Heuer L, Janecek E. *Toxocara* spp. infections in paratenic hosts. Vet Parasitol. 2013; 193: 375–389. <u>https://doi.org/10.1016/j.vetpar.2012.12.033</u> PMID: 23312872
- Epe C, Sabel T, Schnieder T, Stoye M. The behavior and pathogenicity of *Toxocara canis* larvae in mice of different strains. Parasitol Res. 1994; 80: 691–695. PMID: <u>7886040</u>
- 22. Janecek E, Beineke A, Schnieder T, Strube C. Neurotoxocarosis: marked preference of *Toxocara canis* for the cerebrum and *T. cati* for the cerebellum in the paratenic model host mouse. Parasit Vectors. 2014; 7: 194. <u>https://doi.org/10.1186/1756-3305-7-194</u> PMID: 24754900
- Dolinsky ZS, Hardy CA, Burright RG, Donovick PJ. The progression of behavioral and pathological effects of the parasite *Toxocara canis* in the mouse. Physiol Behav. 1985; 35: 33–42. PMID: 4059398
- Heuer L, Beyerbach M, Lühder F, Beineke A, Strube C. Neurotoxocarosis alters myelin protein gene transcription and expression. Parasitol Res. 2015; 114: 2175–2186. <u>https://doi.org/10.1007/s00436-015-4407-1</u> PMID: <u>25773181</u>
- Olson LJ, Rose JE. Effect of *Toxocara canis* infection on the ability of white rats to solve maze problems. Exp Parasitol. 1966; 19: 77–84. PMID: 5925491

- Hamilton CM, Stafford P, Pinelli E, Holland CV. A murine model for cerebral toxocariasis: characterization of host susceptibility and behaviour. Parasitology. 2006; 132: 791–801. <u>https://doi.org/10.1017/</u> S0031182006009887 PMID: 16476184
- Chieffi PP, Aquino RT, Pasqualotti MA, Ribeiro MC, Nasello AG. Behavioral changes in *Rattus norvegi*cus experimentally infected by *Toxocara canis* larvae. Rev Inst Med Trop Sao Paulo. 2010; 52: 243– 246. PMID: 21049227
- Janecek E, Wilk E, Schughart K, Geffers R, Strube C. Microarray gene expression analysis reveals major differences between *Toxocara canis* and *Toxocara cati* neurotoxocarosis and involvment of *T. canis* in lipid biosynthetic processes. Int J Parasitol. 2015; 45: 495–503. <u>https://doi.org/10.1016/j.ijpara.2015.02.009</u> PMID: 25843806
- Fisher M. Toxocara cati: an underestimated zoonotic agent. Trends Parasitol. 2003; 19: 167–170. PMID: <u>12689646</u>
- Jansen J, van Knapen F, Schreurs M, van Wijngaarden T. *Toxocara* ova in parks and sand-boxes in the city of Utrecht. Tijdschr Diergeneeskd. 1993; 118: 611–614. PMID: 8211920
- Fahrion AS, Schnyder M, Wichert B, Deplazes P. *Toxocara* eggs shed by dogs and cats and their molecular and morphometric species-specific identification: Is the finding of *T. cati* eggs shed by dogs of epidemiological relevance? Vet Parasitol. 2010; 177: 186–189. <u>https://doi.org/10.1016/j.vetpar.2010. 11.028 PMID: 21159443</u>
- Fan CK, Holland CV, Loxton K, Barghouth U. Cerebral toxocariasis: silent progression to neurodegenerative disorders? Clin Microbiol Rev. 2015; 28: 663–686. <u>https://doi.org/10.1128/CMR.00106-14</u>
 PMID: <u>26062575</u>
- Irwin S. Comprehensive observational assessment: Ia. A systematic, quantitative procedure for assessing the behavioral and physiologic state of the mouse. Psychopharmacologia. 1968; 13: 222–257. PMID: <u>5679627</u>
- Bouet V, Boulouard M, Toutain J, Divoux D, Bernaudin M, Schumann-Bard P, et al. The adhesive removal test: a sensitive method to assess sensorimotor deficits in mice. Nat Protoc. 2009; 4: 1560– 1564. https://doi.org/10.1038/nprot.2009.125 PMID: 19798088
- Havasiová-Reiterová K, Tomasovicová O, Dubinský P. Effect of various doses of infective *Toxocara canis* and *Toxocara cati* eggs on the humoral response and distribution of larvae in mice. Parasitol Res. 1995; 81: 13–17. PMID: 7724509
- Heuer L, Haendel S, Beineke A, Strube C. Effects of *Toxocara* larvae on brain cell survival by in vitro model assessment. Parasitology. 2015; 142: 1326–1334. <u>https://doi.org/10.1017/S0031182015000694</u> PMID: 26080924
- Hay J, Hutchison WM, Aitken PP. The effect of *Toxocara canis* infection on the behaviour of mice. Ann Trop Med Parasitol. 1983; 77: 543–544. PMID: <u>6660962</u>
- Cox DM, Holland CV. Relationship between three intensity levels of *Toxocara canis* larvae in the brain and effects on exploration, anxiety, learning and memory in the murine host. J Helminthol. 2001; 75: 33–41. PMID: 11316470
- Kayes SG, Omholt PE, Grieve RB. Immune responses of CBA/J mice to graded infections with *Toxo-cara canis*. Infect Immun. 1985; 48: 697–703. PMID: 3997242
- Poulin R. Parasite manipulation of host behavior: an update and frequently asked questions. In: Brockmann J, editor. Advances in the study of behavior. Burlington: Academic Press; 2010. pp. 151–186.
- Hay J, Hutchison WM, Aitken PP. Congenital *Toxoplasma* infection and response to novelty in mice. Ann Trop Med Parasitol. 1983; 77: 437–439. PMID: <u>6639188</u>
- Hutchison WM, Aitken PP, Wells WP. Chronic *Toxoplasma* infections and familiarity-novelty discrimination in the mouse. Ann Trop Med Parasitol. 1980; 74: 145–150. PMID: 7436599
- Flegr J, Kodym P, Tolarová V. Correlation of duration of latent *Toxoplasma gondii* infection with personality changes in women. Biol Psychol. 2000; 53: 57–68. PMID: <u>10876065</u>
- Havlícek J, Gasová ZG, Smith AP, Zvára K, Flegr J. Decrease of psychomotor performance in subjects with latent 'asymptomatic' toxoplasmosis. Parasitology. 2001; 122: 515–520. PMID: <u>11393824</u>
- Chieffi PP, Aquino RT, Paschoalotti MA, Ribeiro MC, Nasello AG. Muscular strength decrease in *Rattus norvegicus* experimentally infected by *Toxocara canis*. Rev Inst Med Trop Sao Paulo. 2009; 51: 73–75. PMID: <u>19390734</u>
- 46. Santos SV, Moura JV, Lescano SA, Castro JM, Ribeiro MC, Chieffi PP. Behavioural changes and muscle strength in *Rattus norvegicus* experimentally infected with *Toxocara cati* and *T. canis*. J Helminthol. 2015; 89: 465–470. https://doi.org/10.1017/S0022149X14000303 PMID: 24725503
- Drago F, Bohus B, Gispen WH, Scapagnini U, de Wied D. Prolactin-enhanced grooming behavior: Interaction with ACTH. Brain Res. 1983; 263: 277–282. PMID: <u>6301648</u>

- Kalueff AV, Stewart AM, Song C, Berridge KC, Graybiel AM, Fentress JC. Neurobiology of rodent selfgrooming and ist value for translational neuroscience. Nat Rev Neurosci. 2016; 17: 45–59. <u>https://doi.org/10.1038/nrn.2015.8</u> PMID: 26675822
- Kim H, Lim CS, Kaang BK. Neuronal mechanisms and circuits underlying repetitive behaviors in mouse models of autism spectrum disorder. Behav Brain Funct. 2016; 12: 3. <u>https://doi.org/10.1186/s12993-016-0087-y PMID: 26790724</u>
- Othman AA, Abdel-Aleem GA, Saied EM, Mayah WW, Elatrash AM. Biochemical and immunopathological changes in experimental neurotoxocariasis. Mol Biochem Parasitol. 2010; 172: 1–8. <u>https://doi.org/10.1016/j.molbiopara.2010.03.006</u> PMID: 20302888
- Bloom FE. Neurotransmission and the central nervous system. In: Hardman JG, Limbind LE, Gilman AG, editors. Goodman and Gilman's the pharmacological basis of therapeutics. 10th ed. USA: McGraw-Hill (Medical Publishing division); 2001.
- Baillieux H, De Smet HJ, Paquier PF, De Deyn PP, Marien P. Cerebellar neurocognition: insights into the bottom of the brain. Clin Neurol Neurosurg. 2008; 110: 763–773. <u>https://doi.org/10.1016/j.clineuro.</u> 2008.05.013 PMID: 18602745
- Morton SM, Bastian AJ. Cerebellar control of balance and locomotion. Neuroscientist. 2004; 10: 247– 259. https://doi.org/10.1177/1073858404263517 PMID: 15155063
- Moser VC. Functional assays for neurotoxicity testing. Toxicol Pathol. 2011; 39: 36–45. <u>https://doi.org/10.1177/0192623310385255</u> PMID: <u>21075917</u>
- Duprez TP, Bigaignon G, Delgrange E, Desfontaines P, Hermans M, Vervoort T, Sindic CJ, Buysschaert M. MRI of cervical cord lesions and their resolution in *Toxocara canis* myelopathy. Neuroradiology. 1996; 38: 792–795. PMID: <u>8957808</u>
- Goffette S, Jeanjean AP, Duprez TP, Bigaignon G, Sindic CJ. Eosinophilic pleocytosis and myelitis related to *Toxocara canis* infection. Eur J Neurol. 2000; 7: 703–706. PMID: <u>11136359</u>
- Kinčeková J, Bánovčin P, Fedor M, Dubinský P Jr., Poláček H, Pavlinová J, Šimeková K. A case of complicated cerebral toxocariasis in a 4-year old child. Helminthologia. 2008; 45: 169–172.
- 58. Hrdá S, Votýpka J, Kodym P, Flegr J. Transient nature of *Toxoplasma gondii*-induced behavioral changes in mice. J Parasitol. 2000; 86: 657–663. <u>https://doi.org/10.1645/0022-3395(2000)086[0657: TNOTGI]2.0.CO;2 PMID: 10958436</u>
- Witting PA. Learning capacity and memory of normal and *Toxoplasma*-infected laboratory rats and mice. Z Parasitenkd. 1979; 61: 29–51. PMID: <u>543216</u>